

LJMU Research Online

Nikiema, AS, Koala, L, Sondo, AK, Post, RJ, Pare, AB, Kafando, CM, Kambire, RS, Sow, B, Bougouma, C, Dabire, RK and Traore, S

The impact of ivermectin on onchocerciasis in villages co-endemic for lymphatic filariasis in an area of onchocerciasis recrudescence in Burkina Faso

http://researchonline.ljmu.ac.uk/id/eprint/17057/

Article

Citation (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Nikiema, AS, Koala, L, Sondo, AK, Post, RJ, Pare, AB, Kafando, CM, Kambire, RS, Sow, B, Bougouma, C, Dabire, RK and Traore, S (2021) The impact of ivermectin on onchocerciasis in villages co-endemic for Ivmphatic filariasis in an area of onchocerciasis recrudescence in Burkina

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact researchonline@ljmu.ac.uk

http://researchonline.ljmu.ac.uk/

http://researchonline.ljmu.ac.uk/



G OPEN ACCESS

Citation: Nikièma AS, Koala L, Sondo AK, Post RJ, Paré AB, Kafando CM, et al. (2021) The impact of ivermectin on onchocerciasis in villages coendemic for lymphatic filariasis in an area of onchocerciasis recrudescence in Burkina Faso. PLoS Negl Trop Dis 15(3): e0009117. https://doi. org/10.1371/journal.pntd.0009117

Editor: Michele Murdoch, Watford General Hospital, UNITED KINGDOM

Received: July 25, 2020

Accepted: January 8, 2021

Published: March 1, 2021

Copyright: © 2021 Nikièma et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>supporting</u> information files.

Funding: The authors received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist. Author Roger Kambire was unable to confirm his authorship contributions. On his behalf, the corresponding

RESEARCH ARTICLE

The impact of ivermectin on onchocerciasis in villages co-endemic for lymphatic filariasis in an area of onchocerciasis recrudescence in Burkina Faso

Achille S. Nikièma¹*, Lassane Koala¹, Apoline K. Sondo², Rory J. Post^{3,4}, Alain B. Paré⁵, Claude M. Kafando⁵, Roger S. Kambiré^{5†}, Bazoumana Sow¹, Clarisse Bougouma⁵, Roch K. Dabiré¹, Soungalo Traoré⁶

1 Institut de Recherche en Sciences de la Santé (IRSS), Direction Régionale de l'Ouest, Bobo Dioulasso, Burkina Faso, 2 Université Ouaga I Pr Joseph ki-Zerbo, Unité de Formation et de Recherche en Sciences de la Santé, Ouagadougou, Burkina Faso, 3 Disease Control Department, London School of Hygiene & Tropical Medicine, London, United Kingdom, 4 School of Biological & Environmental Sciences, Liverpool John Moores University, Liverpool, United Kingdom, 5 Ministère de la Santé, Direction de la Protection de la Santé de la Population, Programme National lutte contre les Maladies Tropicales Négligées, Ouagadougou, Burkina Faso, 6 Medical Entomologist (Freelance), Ouagadougou, Burkina Faso

† Deceased. * achille.nikiema@yahoo.fr

Abstract

In Burkina Faso, onchocerciasis was no longer a public health problem when the WHO Onchocerciasis Control Programme in West Africa closed at the end in 2002. However, epidemiological surveillance carried out from November 2010 to February of 2011, showed a recrudescence of infection in the Cascades Region. This finding was made at a time when ivermectin, a drug recommended for the treatment of both onchocerciasis and lymphatic filariasis, had been distributed in this area since 2004 for the elimination of lymphatic filariasis. It was surprising that ivermectin distributed for treating lymphatic filariasis had not prevented the recrudescence of onchocerciasis. Faced with this situation, the aim of our study was to evaluate the effectiveness of ivermectin on the onchocerciasis parasite. The percentage reduction in microfilarial load after treatment with ivermectin was used as a proxy measure for assessing possible resistance. A cohort study was carried out with 130 individuals who had tested positive for microfilariae of Onchocerca volvulus in 2010 using microscopic examination of skin-snip biopsies from five endemic villages. Subjects were followed from July 2011 to June 2012. The microfilarial load of each individual was enumerated by skinsnip biopsy in 2010, prior to the first ivermectin treatment against onchocerciasis under community guidelines. All individuals received two ivermectin treatments six months apart. In 2012, the microfilarial loads were determined again, six months after the second round of ivermectin and the reductions in parasite loads were calculated to measure the impact of the drug. The percentage reduction of the microfilarial loads ranged from 87% to 98% in the villages. In all villages, there was a statistically significant difference between the average microfilarial loads in 2010 and 2012. The level of reduction of microfilarial loads suggests that ivermectin is effective against the recrudescent population of O. volvulus in Cascades

author has reported his contributions to the best of his knowledge.

Region of Burkina Faso. Further investigations would be necessary to determine the causes of the recrudescence of onchocerciasis. (For French language abstract, see <u>S1 Alternative</u> Language Abstract—Translation of the Abstract into French by the authors.)

Author summary

In 1989, onchocerciasis had been eliminated as a public health problem throughout Burkina Faso by insecticidal treatment of vector breeding sites, but epidemiological surveys along the Comoé River in 2010/11 revealed a recrudescence of infection rates. Modern onchocerciasis control is based upon mass drug administration using ivermectin, and hence biannual distribution of ivermectin was instigated to bring the recrudescence under control. However, it was by no means certain that this was an appropriate strategy because the area was already under mass drug administration with ivermectin since 2004 to eliminate lymphatic filariasis. Onchocerca volvulus adult females with reduced susceptibility to ivermectin have been reported from Ghana, and if the Burkinabe recrudescence was the result of reduced susceptibility, ivermectin might fail to solve the problem. To test this, a cohort of positive people was examined for the density of microfilariae in their skin six months after the second round of ivermectin. The reduction in microfilarial load was substantial, and comparable to levels found in ivermectin-naïve populations (normally assumed to be susceptible), but greater than Ghanaian populations with reduced susceptibility. On this basis we conclude that the Burkinabe recrudescence was not the result of parasites with reduced susceptibility to ivermectin, and biannual treatment remains the chosen means to deal with the recrudescence.

Introduction

Onchocerciasis, or river blindness, is a severely debilitating vector-borne parasitic disease which has been targeted for elimination by the World Health Organization (WHO) by 2025 [1]. It is caused by infection by the filarial nematode worm, *Onchocerca volvulus*, which is transmitted from person to person by blood-feeding blackflies of the genus Simulium (Diptera: Simuliidae). Clinical manifestations of infection include visual impairment (including blindness), skin disease (sub-cutaneous nodules, itching, loss of elasticity and depigmentation) and epilepsy, and as such onchocerciasis is a significant cause of poverty. WHO estimates that 205 million people live in endemic areas worldwide, of whom, more than 99% are in Africa [1,2]. Until the late 1980s there were only a few filaricidal drugs which could be used to treat onchocerciasis and they all had severe side effects. The only means to control onchocerciasis in the community was through vector control, and this was applied from 1974 to 2002 by the WHO Onchocerciasis Control Programme (OCP), which included all endemic areas in Burkina Faso. By 1986 prevalence of microfilariae was less than 5% in all villages in the Comoé river valley in Burkina Faso Cascades Region (except for two, which subsequently dropped to 0% and 3.7% by 1999). OCP ceased vector control operations in this region in 1989 because onchocerciasis was judged to have been reduced to insignificant levels (i.e. less than 5%). However, in 2001 the prevalence in one village in the Comoé river valley had increased to 39.6%, and two more had increased above 5% by 2007. New epidemiological surveys from November 2010 to February 2011 showed that in 13 out of 30 villages in the Comoé river valley prevalence of microfilaremia was above 5% [3]. The reason for the recrudescence is unclear, but it was

associated with high levels of vector infectivity [4] and did not appear to be happening in other parts of southern Burkina Faso [5]. It was particularly worrying because it happened in an area where ivermectin, had been distributed by Mass Drug Administration (MDA) once a year since 2004 for the elimination of lymphatic filariasis (LF) (i.e. for six consecutive years up to 2010). Ivermectin was registered for use against onchocerciasis in 1987, and remains the only drug recommended for the treatment of onchocerciasis through MDA [6], and therefore biannual MDA with ivermectin was introduced to try to bring the recrudescence under control [3].

The main effects of ivermectin are to kill microfilariae and temporarily sterilise the adult female parasite. It was first introduced for the control of onchocerciasis in Africa, but was used successfully for elimination of transmission by the Onchocerciasis Elimination Programme for the Americas (OEPA), and further studies in Africa also showed that onchocerciasis could be eliminated through Community Directed Treatment with Ivermectin (CDTI—a form of MDA) after about 15 years in endemic communities [7,8,9]. Consequently, WHO shifted its objective from control of onchocerciasis as a disease of public health importance to elimination of transmission throughout Africa using CDTI [10]. However, it is potentially problematic that the elimination programme is based on a single suitable drug because of the ever-present threat that the parasite might evolve resistance, and the persistence of microfilaremia was already observed in a number of communities in Ghana after multiple ivermectin treatments [11], and the lack of response in this part of Ghana had persisted up to 2010, and possibly beyond [12].

Considering the high prevalence of onchocerciasis recorded in the 2010/11 survey [3], it seems that the years of ivermectin treatments carried out to eliminate LF have had little or no effect on *O. volvulus*. This situation raised the question whether ivermectin was ineffective on these parasites, and this study aims to test this by presenting the observed changes in microfilarial load in people from five endemic villages in the Cascades Region who tested positive on skin-snip biopsy and took two ivermectin treatments six months apart. The percentage reduction in microfilarial load after ivermectin treatment is used as a proxy measure for assessing possible resistance.

Methods

Ethics statement

The National Onchocerciasis Control Programme received blanket approval from the Ethics Committee of the Ministry of Health in 2009 for carrying out the routine activities for the control and elimination of onchocerciasis (including epidemiological evaluation by skin-biopsy as carried out for this study). In the field on the day of the survey, the team organised an awareness meeting with the village population about the disease and its diagnosis using skin snips. The study was explained to members of the selected communities (in their own language), and those who agreed to participate were selected after free, oral and informed consent in the case of individuals aged 18 and over, and the consent of parents or guardians in the case of those under the age of 18.

Study area and village selection

The study was conducted in the Cascades Health Region, along the Comoé River (which rises in Burkina Faso and flows southwards through Côte d'Ivoire and into the sea). This region is located in the southwest of the country and is irrigated by two rivers (the Comoé and the Léraba) and their tributaries. The region has two seasons, a rainy season from April to October and a dry season throughout the rest of the year. The type of vegetation found in the area is guinea savannah, where vector blackflies of the *Simulium damnosum* complex are generally abundant during the rainy season [13,14]. However, a dam with a capacity of about 38 million cubic meters was built across the River Comoé upstream, near the village of Moussodougou, in 1989, and this dam ensures sufficient flow for the production of blackflies in the rapids located in the study area during the dry season. The Cascades Health Region is subdivided into three Health Districts, Banfora (6295 km²), Mangodara (2808 km²) and Sindou (9302 km²), and the study villages are located in the health districts of Banfora (4 villages) and Mangodara (1 village).

Five villages in the Cascades Region: Badara Karaboro, Badara Nofesso, Bolibana, Congala 2, Kossoumani (Fig 1) were selected for the study. These villages were amongst the worst affected by the recrudescence with unstandardised microfilarial prevalences ranging from 26% to 71% during the 2010 onchocerciasis epidemiological assessment surveys [3]. They are all located between 200 metres (Badara Karaboro) and 2 kilometres (Kossoumani) on either side of the Comoé River. Villages with high prevalences were chosen because it is in these villages that any possible insensitivity of the parasites to ivermectin will be easiest to detect, and also these sorts of villages are most at risk because dermatological and ocular lesions are likely to appear early [15].

Study design

A cohort study of individuals who tested microfilarial positive by skin snip biopsy was established from each of the five study villages. The microfilarial load of each participant was





https://doi.org/10.1371/journal.pntd.0009117.g001

determined by skin-snip biopsy in 2010 before the first round of CDTI in 2011. CDTI was carried out as recommended by the WHO African Programme for Onchocerciasis Control (APOC) [16]. The ivermectin treatment protocol was a single dose of 150 μ g/kg body weight at six-month intervals. The effect of two doses of ivermectin on microfilarial load was estimated by skin-snip six months after the second dose.

Study population

The study population was made up of all the positives from the five villages, making a total of 231 diagnosed individuals. They were followed for 12 months between 2011 and 2012. Their microfilarial load was measured in April 2010, before the first dose of ivermectin (July 2011). The microfilarial load was again measured a second time in June 2012 (six months after the second dose of ivermectin, which was in December 2011, and just before the third dose of ivermectin), from individuals who had taken both ivermectin treatments during the CDTI treatment campaigns (i.e. both in July and December 2011) and who were present on the day of the survey. Of the original 231 people who were skin-snip positive, 101 failed to meet the subsequent requirements of the study for various reasons, and the final size of the cohort was 130 people. The reasons for the 101 people failing to complete the course were varied and included, emigration, death, failure to take one or both doses of ivermectin (as a result of pregnancy or absence at the time of drug distribution) and absence at the time of the final skin-snipping.

Parasitological diagnostic method

The skin-snip biopsy method was used as the diagnostic method according to WHO standards [17]. Skin snips were taken from the right and left iliac crests using a 2-mm Holth corneoscleral punch. These skin snips were then incubated in distilled water for 30 minutes and enumeration of microfilaria was carried out on site. Negative skin ships were then incubated in physiological saline and re-examined 24 hours later. No further skin snips were positive after 24 hours and therefore only results after 30 minutes were used in the analyses of data.

Data analysis

The arithmetic mean of the microfilarial load and the percentage reduction of the arithmetic mean of the microfilarial load were determined. Software R version 3.5.3 was used for the statistical analyses. A Student t Test were used to compare mean microfilarial loads between 2010 and 2012 with a statistical significance level lower than 5%.

Microfilarial load percentage reduction = (C1-C2)/C1*100

C1 = microfilarial load before and C2 = microfilarial load after

Results

The full data set is presented in Supporting Information <u>S1 Table</u>. A total of 130 individuals were examined by skin snip biopsy in 2012, of whom 45% were females. The age extremes were 5 years and 70 years. The maximum individual mean microfilarial load was 115 mf/b (microfilariae per biopsy) in 2010 and 26 mf/b in 2012. The microfilarial loads of 101 people who were skin snip positive in 2010, but dropped out of the study cohort is given in Supporting Information <u>S2 Table</u>.

In comparison with 2010, the microfilarial load was significantly reduced in all villages in 2012 after two treatments with ivermectin (Table 1). In 2012, the mean village load ranged from 0.045 to 2.021 microfilariae per biopsy. Both before and after CDTI, the lowest microfilarial loads were recorded in the village of Kossoumani and the highest load in Bolibana. The

Variables (Village & Sex)	Number examined	2010 Microfilarial load mf/b (sd)	2012 Microfilarial load mf/b (sd)	Reduction (%)	p value
1a. By village					
Badara Karaboro	20	17.425 (16.784)	1.275 (2.658)	92.68	0.0003
Badara Nofesso	26	18.615 (24.722)	1.230 (1.893)	93.39	0.0009
Bolibana	47	18.638 (26.487)	2.021 (4.798)	89.15	0.0004
Congala 2	26	11.346 (8.902)	1.461 (3.042)	87.12	0.00002
Kossoumani	11	2.136 (2.335)	0.045 (0.150)	97.89	0.014
1b. By sex					
Males	72	14.791 (20.720)	1.569 (3.966)	89.39	
Females	58	16.582 (21.720)	1.344 (2.761)	91.89	
Total	130	-			

Table 1. Microfilarial Load of Participants in 2010 and 2012 by Village and by Sex.

Notes: mf/b = mean number of microfilariae per biopsy; sd = Standard deviation.

https://doi.org/10.1371/journal.pntd.0009117.t001

percentage reduction of the mean load varied from 87.1% to 97.9%. The percentage reduction of the microfilarial load was higher in Kossoumani than in the other villages. In both males and females, there was a statistically significant difference between the average loads in 2012 compared to 2010. However, there was no significant difference in the reduction between males and females.

In all age groups, there was a statistically significant difference between the average loads in 2012 compared to 2010 (Table 2). However, there was no significant difference in the reduction between age groups, although the reduction in the microfilarial load was highest in the 50+ age group and lowest in the 20–34 age group.

Discussion

This study was carried out to monitor the response of the recrudescent population of *O. volvulus* to ivermectin treatment which was introduced to eliminate the recrudescence. The recrudescence had taken place in spite of MDA using ivermectin against LF, and it was possible that the recrudescence was enabled by parasite resistance to ivermectin. It is unlikely that any resistance might have been the result of immigration of infected humans carrying ivermectin-tolerant parasites from Ghana (see above) because previous investigations have shown that human infections were not linked to immigration status in the Burkina Faso recrudescence [5], but resistant parasites might still have been carried into the area by migrant vectors, or resistance might have evolved independently in Burkina Faso. This study on the change in the microfilarial load after one year of biannual treatment with ivermectin along the Comoé River was the first of its kind in Cascades Region. Monitoring of the microfilarial load has been done in the

Age group (year)	Number examined	2010 Microfilarial load mf/b (sd)	2012 Microfilarial load mf/b (sd)	Reduction (%)
5-19	60	14.567 (21.248)	1.576 (4.382)	89.18
20-34	31	13.274 (15.610)	1.645 (2.872)	87.60
35-49	24	18.708 (23.190)	1.666 (2.685)	91.09
50+	15	20.400 (28.351)	0.433 (0.677)	97.87
Total	130			

Table 2. Microfilarial Load of Participants in 2010 and 2012 by Age Group.

Notes: mf/b = mean number of microfilariae per biopsy; sd = Standard deviation.

https://doi.org/10.1371/journal.pntd.0009117.t002

past, but as part of the evaluation of the impact of vector control on the epidemiology of onchocerciasis [3].

In this study, evaluation of the effectiveness of ivermectin in reducing microfilarial loads in treated, positive individuals after two rounds of CDTI yielded percentage reductions in all five communities of 87-98% (mean 92%). There have been only four previous studies which recorded the percentage reduction after the second of two biannual treatments, and they recorded the percentage reductions in Liberia (93.8%) [18], Guatemala (mean of 72.2% across three communities) [19] and Sierra Leone (95%) [20]. These reductions were measured in populations which had not been previously exposed to ivermectin and were presumed to be fully susceptible, and our results are comparable. However, in Ghana three communities which had been subject to many annual treatments and were identified as responding poorly to ivermectin (and possibly showing some form of resistance in the parasites) had also been assessed for response to biannual treatment [12]. They showed reductions of 31% (Asubende) 34% (Kyingakrom) and 69% (New Longoro) amongst mf-positive people (mean of three villages = 44.7%). These reductions are clearly much less than those seen from Cascades Region in Burkina Faso and from the ivermectin-naïve communities elsewhere. These comparisons strongly suggests that the recrudescence of onchocerciasis along the Comoé River is not linked to any ineffectiveness of ivermectin against the O. volvulus parasite, and therefore ivermectin can continue to be administered to communities to bring the recrudescence under control and take it to elimination. However, it should be noted that these various studies did not all use the same length of time for incubation of skin snips prior to counting of emergent microfilariae. Our study used the standard method which was adopted by WHO more than 50 years ago [17] but is still being used by WHO [21] and is recommended by WHO along with OV-16 serology for routine monitoring and evaluation surveys [22], with counting of emergent microfilariae after 30 minutes. This was also used in Sierra Leone [20], but the Liberian [18] and Guatemalan [19] studies incubated skin snips "overnight" and for 24 hours respectively. The study of the Ghanaian communities [12] were also incubated for 24 hours. Whilst incubation for only 30 minutes is likely to give an underestimate of community microfilarial load (CMFL) compared with longer periods of incubation, there is no evidence that this will distort the relative reduction in the numbers of microfilariae measured in these studies. In any case, our results are more similar to the results from all three studies of ivermectin-naïve communities, and whilst the reductions of microfilariae in Guatemala are a little less (mean reduction of 72.2%), they are still closer to our results (mean reduction of 92%) than they are to the Ghana results (mean reduction of 44.7%).

The factors which allowed the recrudescence to take place have been discussed in depth by Koala et al. [3], but they are still not fully understood. There are two main questions. Firstly, were the recrudescent parasites the offspring of survivors from the pre-OCP population which had not been eliminated, or were they immigrants carried into the area by infected humans or migrant vectors? Secondly, how could the recrudescence have occurred in the face of mass drug administration of ivermectin (against LF)? When the OCP ceased vector control in 1989, the vectors re-established breeding populations more or less immediately. OCP continued to monitor the annual transmission potential for two years and it never reached zero, but it stayed below the threshold recognised by WHO for self-sustaining transmission [10,22]. However, this was not the case for the epidemiological data provided by skin-snips. WHO no longer considers skin snips to be sufficiently sensitive for the assessment of elimination of transmission although they can be used as a tool to monitor progress towards elimination [22]. However, APOC [10] had considered that the epidemiological transmission threshold had been reached when human prevalence of infection was below 5% in all villages and below 1% in 90% of villages, and communities in the Comoé valley in Burkina Faso had not reached this

threshold when vector control ceased, having 11% of communities above 5% prevalence and 58% of communities above 1% prevalence [3]. Whilst this evidence is far from conclusive, it certainly indicates the possibility that a local population of parasites survived the years of vector control, and the possibility of parasite immigration carried by either vectors or humans appears less likely. This is because the adjacent Leraba River valley was subject to higher levels of vector immigration than the Comoé valley [3] but was not part of the recrudescence, and infections in the recrudescent area seem to be associated with the resident population rather than those with a migration history [5]. Whether the recrudescent parasite population consisted of local survivors or immigrants, the question still remains as to why it was able to recrudesce in the face of on-going ivermectin distribution against LF. We have now shown that the recrudescent parasites do not seem to be resistant to ivermectin, and so the most likely explanation is an inadequacy of the MDA activities in the Cascades Region by the National Programme for the Eradication of LF. Koala et al. [3] mentioned that during investigations "it became evident that therapeutic coverage had often fallen short. Inhabitants gave a number of reasons for non-participation, including the undesirable effects of the drugs ... and a perception that LF was not a grave problem. Furthermore, the programme achieved only weak geographic coverage in these zones (with certain villages untreated)" (see S3 Table), and Nikièma [23] found that the implementation of the CDTI strategy against onchocerciasis in 2011 was better monitored through the treatment register and yielded reliable, valid and satisfactory therapeutic and geographical coverage rates. In conclusion, it seems most likely that a local population of parasites had survived the years of vector control and were able to recrudesce in the face of mass drug administration of ivermectin against LF because the coverage was inadequate. However, further investigations would be needed to determine the causes of the recrudescence with any certainty.

Whilst other studies have shown tolerance (or even resistance) of ivermectin by *O. volvulus* in Ghana [24], our results indicate that this was not the case in Burkina Faso, but the recrudescence was still dangerous. Our results show that in 2010, the mean of the microfilarial load was greater than 2 microfilariae/biopsy (mf/b) in all the five villages, and it has been shown that *S. damnosum* s.l. vectors were able to ensure the transmission of the parasite in forest regions of Côte d'Ivoire at 1 mf/b [25]. These positive individuals would therefore constitute a risk to the rest of the community. Furthermore, the parasite strain in the area is expected to be the savannah strain (which causes high rates of blindness), and the vectors are expected to belong to the 'savannah group' of *S. damnosum* s.l. cytospecies which are effective vectors of blinding onchocerciasis [4,26,27].

The reduction of the microfilarial load brought about by CDTI is expected to have an impact on the transmission in the villages. The numbers of microfilaria taken up by the black-flies during their blood-meal is reduced with the corollary that fewer microfilaria can cross the peritrophic membrane to develop into infective larvae [26,28]. It is the infective larvae that are transmitted by the blackfly when it takes its blood-meal. Studies carried out during an outbreak in the Guinea savannah in Ghana showed a 65–85% reduction in transmission of the parasite by savannah species of the *S. damnosum* complex during the three months following ivermectin mass distribution [29]. Other data remains to be collected in our study, specifically entomological data, but we could expect that biannual CDTI will reduce the transmission of onchocerciasis in the Comoé valley. Finally, the reduction of the individual microfilarial load from a maximum of 115 mf/b in 2010 to a maximum of 26 mf/b in 2012 should be of benefit to the individual patient. This individual load reduction is characterized as a "curative effect" by some authors [30]. This study did not investigate the dermatological and ophthalmological aspects of onchocerciasis, but the improvement of important symptoms (skin and eye lesions) after administration of ivermectin in onchocerciasis endemic areas in sub-Saharan Africa has

been widely reported [31,32,33]. The age group of 50 and over had the highest percentage reduction of microfilariae after ivermectin administration in Cameroon [34]. This might be partly explained by older individuals being carriers of older adult female macrofilariae, which would release fewer microfilariae after ivermectin treatment, as opposed to younger individuals who would be carriers of younger female macrofilariae, which are able to produce more microfilaria after ivermectin treatment.

The effectiveness of ivermectin observed on the parasite in this study suggested that by continuing the CDTI in the region (and maintaining treatment coverage of 80% or higher, with an emphasis on close supervision by health professional and community distributors), it should be possible to eliminate transmission and all the consequent benefits would be felt by the population, as demonstrated by previous studies showing that eliminating onchocerciasis is feasible through regular treatment with ivermectin [7,8,9]. Therefore, the biannual treatments with ivermectin introduced in 2011 have been continued. An epidemiological (skin-snip) survey carried out in 2016 indicated a mean reduction of mf prevalence of around 75% and a second survey is planned for 2021, and the results will be published when the recrudescence is eliminated.

Supporting information

S1 Alternative Language Abstract. Translation of the Abstract into French by the authors. (PDF)

S1 Table. Numbers of microfilariae for each iliac crest (mfs/b) in all participants in 2010 and 2012.

(PDF)

S2 Table. Numbers of microfilariae for each iliac crest (mfs/b) in the 101 people who were skin snip positive in 2010 but failed to complete the cohort requirements to participate in the skin snip assessment in 2012. (PDF)

S3 Table. Therapeutic coverage of mass drug administration with ivermectin against lymphatic filariasis in the onchocerciasis recrudescence area of Cascades Region in Burkina Faso up to 2011.

(PDF)

Acknowledgments

This work was carried out as part of the onchocerciasis elimination activities of the National NTD Control Programme of the Ministry of Health, which is assisted by SightSavers. We are grateful to the Regional Director of Health of the Cascades and his staff for the material support and the mobilization of community leaders. We also thank the community distributors of the study villages who helped the mobilization of the study participants. Finally, we would like to express our gratitude to all the inhabitants of the study villages for their participation to this study and their kindness.

Author Contributions

Conceptualization: Achille S. Nikièma, Roger S. Kambiré, Soungalo Traoré. **Data curation:** Achille S. Nikièma, Alain B. Paré, Bazoumana Sow, Soungalo Traoré. **Formal analysis:** Achille S. Nikièma, Alain B. Paré, Bazoumana Sow, Soungalo Traoré. Funding acquisition: Apoline K. Sondo, Roch K. Dabiré, Soungalo Traoré.

Investigation: Achille S. Nikièma, Lassane Koala, Alain B. Paré, Claude M. Kafando, Roger S. Kambiré, Soungalo Traoré.

Methodology: Achille S. Nikièma, Roger S. Kambiré, Soungalo Traoré.

Project administration: Soungalo Traoré.

Supervision: Rory J. Post, Roger S. Kambiré, Roch K. Dabiré, Soungalo Traoré.

Validation: Achille S. Nikièma, Roger S. Kambiré, Soungalo Traoré.

Visualization: Achille S. Nikièma, Rory J. Post, Roger S. Kambiré, Soungalo Traoré.

Writing - original draft: Achille S. Nikièma, Roger S. Kambiré, Soungalo Traoré.

Writing – review & editing: Achille S. Nikièma, Lassane Koala, Apoline K. Sondo, Rory J. Post, Alain B. Paré, Claude M. Kafando, Roger S. Kambiré, Bazoumana Sow, Clarisse Bougouma, Roch K. Dabiré, Soungalo Traoré.

References

- Dadzie Y, Amazigo U, Boatin A, Sékétéli A. Is onchocerciasis elimination in Africa feasible by 2025: a perspective based on lessons learnt from the African control programmes. Infect Dis Poverty. 2018; 7:63. https://doi.org/10.1186/s40249-018-0446-z PMID: 29966535
- WHO. Progress report on the elimination of human onchocerciasis, 2017–2018. Wkly Epidemiol Rec. 2018; 93:633–648. Available from: https://www.who.int/wer/2018/wer9347/en/.
- Koala L, Nikiema A, Post R, Paré A, Kafando C, Drabo F, Traoré S. Recrudescence of onchocerciasis in the Comoé valley in southwest Burkina Faso. Acta Trop. 2017; 166:96–105. https://doi.org/10.1016/j. actatropica.2016.11.003 PMID: 27845063
- Koala L, Nikièma A, Paré A, Drabo F, Toé L, Belem A, et al. Entomological assessment of the transmission following recrudescence of onchocerciasis in the Comoé Valley, Burkina Faso. Parasit Vectors. 2019; 12:34. https://doi.org/10.1186/s13071-019-3290-5 PMID: 30646934
- Nikièma A, Koala L, Post R, Paré A, Kafando C, Drabo F, et al. Onchocerciasis prevalence, human migration and risks for onchocerciasis elimination in the Upper Mouhoun, Nakambé and Nazinon river basins in Burkina Faso. Acta Trop. 2018; 185:176–182. https://doi.org/10.1016/j.actatropica.2018.05. 013 PMID: 29782820
- 6. Thylefors B. The Mectizan Donation Program (MDP). Ann Trop Med Parasitol. 2008; 102:S39–S44. https://doi.org/10.1179/136485908X337481 PMID: 18718154
- Diawara L, Traoré M, Badji A, Bissan Y, Doumbia K, Goita S, et al. Feasibility of onchocerciasis elimination with ivermectin treatment in endemic foci in Africa: first evidence from studies in Mali and Senegal. PloS Negl Trop Dis. 2009; 3:e497. https://doi.org/10.1371/journal.pntd.0000497 PMID: 19621091
- Traoré M, Sarr M, Badji A, Bissan Y, Diawara L, Doumbia K, et al. Proof-of-principle of onchocerciasis elimination with ivermectin treatment in endemic foci in Africa: final results of a study in Mali and Senegal. PloS Negl Trop Dis. 2012; 6:e1825. https://doi.org/10.1371/journal.pntd.0001825 PMID: 23029586
- Tekle A, Elhassan E, Isiyaku S, Amazigo U, Bush S, Noma M, et al. Impact of long treatment of onchocerciasis with ivermectin in Kaduna state, Nigeria: first evidence for elimination in the operational area of the African Programme for Onchocerciasis Control. Parasit Vectors. 2012; 5:28. <u>https://doi.org/10. 1186/1756-3305-5-28 PMID: 22313631</u>
- APOC. Conceptual and Operational Framework of Onchocerciasis Elimination with Ivermectin Treatment. Ouagadougou: World Health Organisation African Programme for Onchocerciasis Control. WHO/APOC/MG/10.1, 2010. Available from: https://apps.who.int/iris/bitstream/handle/10665/275466/ JAF16.6-II-eng.pdf.
- Awadzi K, Boakye D, Edwards G, Opoku N, Attah S, Osei-Atweneboana M, et al. An investigation of persistent microfilaridermias despite multiple treatments with ivermectin, in two onchocerciasisendemic foci in Ghana. Ann Trop Med Parasitol. 2004; 98:231–249. https://doi.org/10.1179/ 000349804225003253 PMID: 15119969
- 12. Frempong K, Walker M, Cheke R, Tetevi E, Gyan E, Owusu E, et al. Does increasing treatment frequency address suboptimal responses to ivermectin for the control and elimination of River Blindness? Clin Infect Dis. 2016; 62:1338–1347. https://doi.org/10.1093/cid/ciw144 PMID: 27001801

- Le Berre R. Contribution à l'étude biologique et écologique de Simulium damnosum Théobald, 1903 (Diptera-Simuliidae). Paris: Mémoires ORSTOM 1966: 17 Available from: http://horizon. documentation.ird.fr/exl-doc/pleins_textes/pleins_textes_2/memoires/10961.pdf.
- Philippon B.1977. Étude de la transmission d'Onchocerca volvulus (Leuckart, 1893) (Nematoda, Onchocercidae) par Simulium damnosum Théobald, 1903 (Diptera, Simulidae) en Afrique Tropicale. Paris: Travaux et Documents de l'ORSTOM 1977; 6. Available from: <u>http://horizon.documentation.ird.fr/exldoc/pleins_textes/divers09-10/08626.pdf</u>.
- Dadzie K, Remme J, Rolland A, Thylefors B. The effect of 7–8 years of the vector control on the evolution of ocular onchocerciasis in West African savannah. Trop Med Parasitol. 1986; 37:263–270. PMID: 3787122
- APOC/WHO. Community-Directed Treatment with Ivermectin (CDTI): A Practical Guide for Trainers of Community-Directed Distributors. Ouagadougou: African Programme for Onchocerciasis Control; 1998. Available from: https://www.who.int/apoc/publications/guidefortrainers_cdds/en/.
- Prost A, Prod'hon J. Le diagnostic parasitologique de l'onchocercose. Revue critique des méthodes en usage. Med Trop (Mars). 1978; 38: 519–532. Available from: https://core.ac.uk/download/pdf/ 39879244.pdf.
- Greene B, Dukuly Z, Munoz B, White A, Pacque M, Taylor H. A comparison of 6-, 12-, and 24-monthly dosing with ivermectin for treatment of onchocerciasis. J Infect Dis. 1991; 163:376–380. https://doi.org/ 10.1093/infdis/163.2.376 PMID: 1988521
- Collins R, Gonzales-Peralta C, Castro J, Zea-Flores G, Cupp M, Richards F Jr, et al. Ivermectin: Reduction in prevalence and infection intensity of Onchocerca volvulus following biannual treatments in five Guatemalan communities. Am J Trop Med Hyg. 1992; 47:156–169. https://doi.org/10.4269/ajtmh.1992. 47.156 PMID: 1503185
- Whitworth J, Maude G, Downham M. Clinical and parasitological responses after up to 6.5 years of ivermectin treatment for onchocerciasis. Trop Med Int Hlth 1996; 1:786–793. <u>https://doi.org/10.1111/j.</u> 1365-3156.1996.tb00111.x PMID: 8980590
- Tekle A, Elhassan E, Isiyaku S, Amazigo U, Bush S, Noma M, et al. Impact of long-term treatment of onchocerciasis with ivermectin in Kaduna State, Nigeria: first evidence of the potential for elimination in the operational area of the African Programme for Onchocerciasis Control. Parasit Vectors 2012; 5:28. Available from: https://link.springer.com/article/10.1186/1756-3305-5-28. https://doi.org/10.1186/1756-3305-5-28 PMID: 22313631
- WHO. Guidelines for Stopping Mass Drug Administration and Verifying Elimination of Human Onchocerciasis. Geneva: WHO document WHO/NTD/PCT/2016.1, 2016. Available from: https://www.who.int/neglected_diseases/resources/9789241510011/en/.
- 23. Nikièma S. Contribution à la lutte contre l'onchocercose au Burkina Faso: évaluation de l'efficacité en 2011–2012 du Traitement à l'Ivermectine sous Directives Communautaires (TIDC) dans la Région des Cascades. Thèse de Pharmacie, Université de Ouagadougou. 2013.
- Osei-Atweneboana M, Eng J, Boakye D, Gyapong J, Prichard R. Prevalence and intensity off Onchocerca volvulus infection and efficacy of ivermectin in endemic communities in Ghana: a two-phase epidemiological study. Lancet. 2007; 369:2021–2029. <u>https://doi.org/10.1016/S0140-6736(07)60942-8</u> PMID: 17574093
- Brika G. Étude du Potentiel infestant des Onchocerquiens en fonction de leurs Charges Microfilariennes. Zones de grandes rivières de forêt de Côte d'Ivoire. Mémoire de Diplôme d'Études Approfondies, Université de Bouaké. 1989.
- Philippon B, Bain O. Transmission de l'onchocercose humaine en zone de savane d'Afrique Occidentale. Passage des microfilaires d'*Onchocerca volvulus* Leuck. dans l'hémocèle de la femelle *Simulium damnosum* Théobald. Cahiers O.R.S.T.O.M. Série entomologie médicale et parasitologie. 1972; 10:251–61. Available from: http://horizon.documentation.ird.fr/exl-doc/pleins_textes/cahiers/entomo/ 18971.pdf.
- Toé L, Koala L, Burkett-Cadena N, Traoré B, Sanfo M, Kambiré S, et al. Optimization of the Esperanza window trap for the collection of the African onchocerciasis vector Simulium damnosum s.l. Acta Trop. 2014; 39:1–5. https://doi.org/10.1016/j.actatropica.2014.04.029 PMID: 24794201
- Quillévéré D. Contribution à l'étude des caractéristiques taxonomiques, bioécologiques et vectrices des membres du complexe *Simulium damnosum* présents en Côte d'Ivoire. Paris: Travaux et Documents de l'ORSTOM. 1979; 109. Available from: http://horizon.documentation.ird.fr/exl-doc/pleins_textes/ divers09-10/09869.pdf.
- Remme J, Baker R, De Sole G, Dadzie K, Walsh J, Adams M, et al. A community trial of ivermectin in the onchocerciasis focus of Asubende, Ghana. Effect on the microfilarial reservoir and the transmission of *Onchocerca volvulus*. Trop Med Parasitol. 1989; 40:367–374. PMID: 2617046

- Boussinesq M. Contribution du MECTIZAN au contrôle de l'onchocercose: Impact sur la transmission après huit années de traitement. Sante. 1998; 8:20–24. Available from: http://horizon.documentation. ird.fr/exl-doc/pleins_textes/divers19-08/010043771.pdf. PMID: 9592871
- Prod'hon J, Boussinesq M, Fobi G, Prud'hom J, Enyong P, Lafleur C, et al. Lutte contre l'onchocercose par l'ivermectine: résultats d'une campagne de masse au nord-Cameroun. Bull World Health Organ, 1991; 69:443–450. Available from: https://apps.who.int/iris/bitstream/handle/10665/261429/ PMC2393233.pdf. PMID: 1934238
- 32. Resnikoff S. Contribution du MECTIZAN au contrôle de l'onchocercose: impact sur les lésions oculaires. Sante. 1998; 8:12–14. PMID: 9592869
- Ozoh G, Murdoch M, Bissek A-C, Hagan M, Ogbuagu M, Shamad M, et al. The African Programme for Onchocerciasis Control: impact on onchocercal skin disease. Trop Med Int Health. 2011; 16:875–883. https://doi.org/10.1111/j.1365-3156.2011.02783.x PMID: 21481109
- Boussinesq M, Prod'hon J, Chippaux J, Quillevéré D. Effet à long terme d'une prise unique d'ivermectine sur les charges microfilariennes dermiques dans un foyer d'onchocercose du nord-Cameroun. Ann Soc Belg Med Trop. 1993; 73:189–196. Available from: http://lib.itg.be/open/ASBMT/1993/ 1993asbm0189.pdf. PMID: 8279866